Author's response to reviews

Title: Impact on mortality and cancer incidence rates of using random invitation from population registers for recruitment to trials

Authors:

Matthew Burnell (m.burnell@ucl.ac.uk)
Aleksandra Gentry-Maharaj (a.gentry-maharaj@ucl.ac.uk)
Andy Ryan (a.ryan@ucl.ac.uk)
Sophia Apostolidou (s.apostolidou@ucl.ac.uk)
Mariam Habib (m.habib@ucl.ac.uk)
Jatinderpal Kalsi (j.k.kalsi@ucl.ac.uk)
Steven Skates (sskates@partners.org)
Mahesh Parmar (max.parmar@ctu.mrc.ac.uk)
Mourad W Seif (mseif@manchester.ac.uk)
Nazar N Amso (amsonn@Cardiff.ac.uk)
Keith Godfrey (keith.godfrey@ghnt.nhs.uk)
David Oram (david.oram@bartsandthelondon.nhs.uk)
Jonathan Herod (jonathan.herod@lwh.nhs.uk)
Karim Williamson (Karim.Williamson@nuh.nhs.uk)
Howard Jenkins (Howard.Jenkins@derbyhospitals.nhs.uk)
Tim Mould (tim.mould@uclh.org)
Robert Woolas (r.woolas@btinternet.com)
John Murdoch (j.murdoch@UHBristol.nhs.uk)
Stephen Dobbs (Stephen.Dobbs@bch.n-i.nhs.uk)
Simon Leeson (Simon.Leeson@nww-tr.wales.nhs.uk)
Derek Cruikshank (Derek.Cruickshank@stees.nhs.uk)
Stuart Campbell (profscampbell@hotmail.com)
Lesley Fallowfield (l.j.fallowfield@sussex.ac.uk)
Ian Jacobs (i.jacobs@ucl.ac.uk)
Usha Menon (u.menon@ucl.ac.uk)

Version: 6 Date: 22 February 2011

Author's response to reviews:

The revisions are just some very minor alterations to the text, such as a couple of typos, space deletions, some sentence rewording as well as minor changes needed following formatting by Trials team (a couple of numbers in table were unnecessarily boldened, two extra horizontal lines re-inserted into table 4)